## **Author's Response To Reviewer Comments**

Clo<u>s</u>e

As suggested by reviewer 2, we have corrected the typos in the text. We would also like to include a Research Object bundle for our CWL workflow as a supplemental file; however the EM system unzips the RO bundle. I'll be happy to submit it by email.

Clo<u>s</u>e